Author’s response to reviews

Title: Nicotine dependence in Croatian male inpatients with schizophrenia

Authors:
Marina Sagud (marinasagud@mail.com)
Bjanka Vuksan-Cusa (bjanka.vuksan@gmail.com)
Nenad Jaksic (nenad_jaksic@yahoo.com)
Alma Mihaljevic Peles (apeles@mef.hr)
Maja Zivkovic (maja.zivkovic@zg.t-com.hr)
Suzana Vlatkovic (suzana.vlatkovic@gmail.com)
Tea Prgic (tea.prgic1985@gmail.com)
Darko Marcinko (darko.marcinko@zg.t-com.hr)
Wei Wang (drwangwei@zju.edu.cn)

Version: 4 Date: 10 Jan 2018

Author’s response to reviews:

Date: January 10th, 2018.

To: "BMC Psychiatry - Editorial Office"

From: "Marina Sagud" marinasagud@mail.com

Subject: Submission to BMC Psychiatry - BPSY-D-17-00582R3

Dear Dr. Hickey,

Thank you very much for sending us additional corrections of our manuscript "Nicotine dependence in Croatian male inpatients with schizophrenia" (BPSY-D-17-00582R3). We also want to express our gratitude for your efforts and patience regarding our article.

We have now addressed all Editor's comments, which are mentioned below;
1. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript. Thank you for this correction. "Not applicable" was stated in "Consent for publication" section, instead of previous "Participants were informed about plans for publication, all information is anonymous"- which was deleted.

2. We note that Darko Marcinko and Wei Wang have not been mentioned in Authors' contributions. Please ensure all authors are included in this section, along with their individual contributions. Thank you for this correction. In the part of the text: "all authors actively participated in the interpretation of data and have read and approved the final version of the manuscript"; "…all authors…” was deleted and: “ while MŠ, BVĆ, NJ, AMP, MŽ, SV, TP, DM and WW” was added. The revised statement is now: “while MŠ, BVĆ, NJ, AMP, MŽ, SV, TP, DM and WW actively participated in the interpretation of data and have read and approved the final version of the manuscript”.

According to the instructions, the manuscript is provided as a single clean copy without any tracked changes.

Best regards,

Marina Šagud